Author’s response to reviews

Title: Splicing analysis of CYP11B1 mutation in a family affected with 11β-hydroxylase deficiency: case report

Authors:
Pattaranatcha Charnwichai (gao_gaabgab@hotmail.com)
Patra Yeetong (patra163@hotmail.com)
Kanya Suphapeetiporn (kanya.su@chula.ac.th)
Vichit Supornsilchai (pedploy@hotmail.com)
Taninee Sahakitrungruang (ptaninee@hotmail.com)
Vorasuk Shotelersuk (vorasuk.s@chula.ac.th)

Version: 2 Date: 30 May 2016

Author’s response to reviews:

May 30, 2016
Dr. Philippa Harris
Editor-in-Chief
BMC Endocrine Disorders

Dear Dr. Harris,

Thank you for your email of May 20, 2016 indicating that our manuscript: BEND-D-16-00038R1 entitled “Splicing analysis of CYP11B1 mutation in a family affected with 11β-hydroxylase deficiency: case report” will be acceptable for publication once a few editorial
requests have been addressed. We offer the following clarifications and changes in response to the editorial requests as highlighted in yellow.

Comments:

Reviewer #1:
The Authors have suitably addressed this reviewers minor recommendations and the manuscript is now suitable for publication.

We thank the Reviewer for his interest in our work.

-------------------

Editorial Requests

* Please remove the CARE checklist from the additional files

We have removed the CARE checklist as suggested.

* Please ensure that your manuscript contains a Declarations section before the references, with ALL of the following subsections:
  
  * List of abbreviations – provided on page 10, line 219-223.

  * Ethics approval and consent to participate – provided on page 11, line 224-227.

  * Consent for publication – as you are not publishing identifying patient data, you can write "Not applicable" in this section – provided on page 11, line 229.

  * Availability of data and materials – please write 'All data contained within the article' here – provided on page 11, line 231.
We thank the reviewers and editors for their enthusiasm for our work. We really appreciate all your comments and trust that the revised manuscript will now be acceptable for publication in “BMC Endocrine Disorders”.

Sincerely,
Taninee Sahakitrungruang, M.D.
Associate Professor of Pediatrics
Division of Pediatric Endocrinology, Department of Pediatrics, Faculty of Medicine
Chulalongkorn University, Bangkok, Thailand 10330
Phone +6681-646-8083, Fax +662-256-4911
E-mail: Taninee.P@chula.ac.th, ptaninee@hotmail.com